
    Technical Infrastructure (TI) 

Facility Coordinator last week Gildas Langlois 

Facility Coordinator this week Ronan Ledru 

Statistics 

Alarms  

Phone calls 1393 

ODMs 186 

Facility Status 

Summary  

Issues 

Tuesday December 17th at 02:10 
Emergency stop on BEQ1 
Firefighters on site but nothing found 
After investigation from EN-EL, the button was full of water. 
The chain has been opened when the water has frozen 
 
Wednesday December 18th 

Grey water leak in building 5 
Fire brigade and SCE on site to fix the issue 
Lift out of order because of the leak 
 
Friday Decembre 20th at 9h20 
Fire on UPS EBS11/82 in RE82 
Trip of EOD5/82 (Cryo and IT impacted) 
A Borri technician was already at CERN, so the UPS has been repaired during 
the day. 
 
Sunday Decembre 22th 
A tent has blow away from the traveller camp in the BE2 substation 
Fire brigade onsite for taking out the tent 
 
 

Plans  

 


